Author’s response to reviews

Title: The KTX360°-study: A multicenter, multisectorial, multimodal, telemedicine-based followup care model to improve care and reduce health-care costs after kidney transplantation in children and adults

Authors:
Lars Pape (Pape.Lars@mh-hannover.de)
Martina de Zwaan (deZwaan.Martina@mh-hannover.de)
Uwe Tegtbur (Tegtbur.Uwe@mh-hannover.de)
Fabiola Feldhaus (Feldhaus.Fabiola@mh-hannover.de)
Julia Wolff (Julia.Wolff@iges.com)
Lena Schiffer (Schiffer.Lena@mh-hannover.de)
Christian Lerch (Lerch.Christian@mh-hannover.de)
Nils Hellrung (n.hellrung@symeda.de)
Volker Kliem (v.kliem@khmue.de)
Gerhard Loennemann (lonnemann@eickenhof-dialyse.de)
Hans-Dieter Nolting (Hans-DieHans-Dieter.Nolting@iges.com)
Mario Schiffer (Schiffer.Mario@mh-hannover.de)

Version: 2 Date: 12 Aug 2017

Author’s response to reviewers
Answers to the Editor Comments:

Was the trial retrospectively registered? In the ISRCTN file it states that it was. If so, you need to state retrospectively registered after the trial number in the Abstract (please ensure you add this to the Abstract of the Submission system as well).

The recruitment for the trial started on 2nd of May 2017 and the registration was finally done on 5th of May 2017. As this was three days later I think that “retrospectively” is right. I have
therefore added “retrospectively registered” to the abstract in the manuscript and in the submission system.

Please change the title of the Ethics approval section to Ethics approval and consent to participate
Done

Please remove the additional Figure 1 from the main text.
Done